Author's response to reviews

Title: Effect of Working Characteristics and Taught Ergonomics on The Prevalence of Musculoskeletal Disorders Amongst Dental Students

Authors:

Saad A Khan (saad_khan@imu.edu.my)
Kwai Yee Chew (jamiecky29@yahoo.com)

Version: 5 Date: 9 March 2013

Author's response to reviews:

Dear Editor,

The Manuscript has been edited by a Senior Professor whose native language is English. The new words added are in font red while the words which are removed are bracketed.

Thank you for patience.

Best regards,

SA Khan